Author's response to reviews

Title: Cost-effectiveness of Collaborative Care including PST and an Antidepressant Treatment Algorithm for the treatment of Major Depressive Disorder in primary care; a Randomised Clinical Trial.

Authors:

Marjoliek A IJff (mijff@trimbos.nl)
Klaas ML Huijbregts (khuijbregts@trimbos.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Aartjan TF Beekman (aartjanb@ggzba.nl)
Leona Hakkaart-van Roijen (l.hakkaart-vanroijen@erasasmusmc.nl)
Frans Rutten (f.rutten@erasasmusmc.nl)
Jurgen Unutzer (unutzer@u.washington.edu)
Christina M van der Feltz-Cornelis (cfeltz@trimbos.nl)

Version: 4 Date: 12 January 2007

Author's response to reviews: see over
Dear editor,

I have resubmitted my manuscript ‘Cost-effectiveness of Collaborative Care including PST and an Antidepressant Treatment Algorithm for the treatment of Major Depressive Disorder in primary care; a Randomised Clinical Trial’ (Manuscript ID 1945482116126357), since I have added another author.

The only things changed in the manuscript are therefore the ‘acknowledgements’-section and the ‘authors contributions’-section.

If I can provide you with any other information, please do not hesitate to contact me.

Regards,

Marjoliek IJff